Author's response to reviews

Title: Prenatal diagnosis of a rare form of congenital mid-ureteral stricture: a case report and literature revisited

Authors:

Milena Brugnara (brugnara@supereva.it)
Mariangela Cecchetto (mariangela.cecchetto@libero.it)
Riccardo Manfredi (riccardo.manfredi@univr.it)
Zuffante Michele (michelezuffante@yahoo.it)
VassilioS Fanos (vfanos@tiscali.it)
Marco Zaffanello (marco.zaffanello@univr.it)
Angelo Pietrobelli (angpie@tin.it)

Version: 2 Date: 15 February 2007

Dear Senior Assistant Editor
Dr Annabel Philips

Please find the revised version of the manuscript. The revision process has been made in agreement with the reviewer's suggestions.

Many thanks for the suggestions made by BK. In agreement, we revised the grammar writing of the manuscript (by AP).

The figures were modified in agreement with the suggestions by the reviewer GK.
GFR is normal and recently confirmed again.
The urography of the Figure 2 (ex figure 3) panel C is an intrasurgical retrograde ureteropyelogram.
In agreement with pediatric urologists the sentence "remodelling of the renal pelvis..." was deleted because not correct. Than you for the correction.

Best regards